Author's response to reviews

Title: Anti-thrombotic effects of alpha-linolenic acid isolated from Zanthoxylum bungeanum Maxim seeds

Authors:

Qian Yang (yangqiangqfm@163.com)
Weidong Cao (weidcao@126.com)
Xuanxuan Zhou (xuanxuanzhou@hotmail.com)
Wei Cao (weicaomu@yeah.net)
Yanhua Xie (yhxiefmmu@163.com)
Siwang Wang (siwwang@126.com)

Version: 5 Date: 31 August 2014

Author's response to reviews:

Editors' Comments:
Authors have addressed most of the concerns previously raised, but few errors still to be corrected
- ?Zanthoxylum bungeanum? is a species name, it should be in italic along the MS

Response: Above points have been corrected in the revised manuscript.